

## COLLEGIUM OF ECONOMIC ANALYSIS WORKING PAPER SERIES

# GDP Effects of Pandemics: A Historical Perspective

Maciej Stefański

## GDP Effects of Pandemics: A Historical Perspective

Maciej Stefański SGH Warsaw School of Economics<sup>1</sup>

December 2020

#### Abstract

The paper estimates dynamic effects of pandemics on GDP per capita with local projections, controlling for the effects of wars and weather conditions, using a novel dataset that covers 33 countries and stretches back to the 13th century. Pandemics are found to have prolonged and highly statistically significant effects on GDP per capita - a pandemic killing 1% of the population tends to increase GDP per capita by approx. 0.3% after about 20 years. The results are qualitatively robust to various model specifications, geographical division of the sample and an exclusion of extreme events such as the Black Death and the New World epidemics. The effects of pandemics differ from those of wars and weather, which are negative and die out quicker, in line with the neoclassical growth model.

Keywords: pandemic, GDP, local projection, economic history, war, tree rings.

JEL: I15, N10, N30, N40, N50, O47.

<sup>&</sup>lt;sup>1</sup> The views expressed in this paper belong to the author only, and have not been endorsed by SGH or any other institutions. I am grateful to Adam Pigoń for useful suggestions. E-mail address: <u>maciej.stefanski@doktorant.sgh.waw.pl</u>

## 1. Introduction

The COVID-19 pandemic has reignited interest in the study of macroeconomic effects of epidemics. Will the effects of the pandemic be short-lasting, with the economy returning to its pre-crisis trend sooner or (just a bit) later, or will they last much longer, with GDP remaining depressed for many years or even decades to come? Perhaps history can help us answer this question to some degree – despite all the differences in the level of economic, organisational and healthcare development between the present and the past.

From a theoretical viewpoint, past pandemics may have had either a positive or negative impact on GDP per capita. On the one hand, they decreased the denominator, i.e. the population size, which could have led to an increase in the standard of living in Malthusian subsistence societies (Malthus 1798). In a more sophisticated economy, they also tended to increase the capital labour ratio, rising labour productivity and per capita income, in line with the neoclassical growth model (Solow 1956). A large catastrophic event like a pandemic may have also disturbed economic and power structures, perhaps leading to positive institutional changes and technological innovation, in the spirit of Schumpeterian creative destruction (Schumpeter 1942).

On the other hand, in the short run pandemics were likely to raise uncertainty and induce precautionary saving, lowering investment and consumption even without lockdowns and quarantines. The level of risk aversion may as well have increased in a more permanent manner as economic agents would fear the repetition of the catastrophe, in a mechanism similar to Bayesian updating (Kozlowski et al. 2020). At the same time, lower population density following a pandemic could have led to a decline in market economy activities (surplus production and trade) in a longer run due to lower effects of scale and higher transportation and transaction costs.

The identification of GDP per capita effects of pandemics is therefore very much an empirical question. To answer it, data on the size of pandemics (in terms of percentage of population killed by each pandemic) is assembled for a panel of 33 countries between 1252 and 2016. Since pandemics are not necessarily fully exogenous - the historical evidence suggests they might have been fostered by wars, droughts and poor living conditions – data on tree ring growth (to control for weather conditions) and incidence of wars is also gathered. Other than controlling for an endogenous element of pandemics, this allows me to compare the GDP effects of pandemics with those of wars and weather conditions.

After combining the gathered data with GDP per capita estimates from the Maddison Project Database, local projections in the spirit of Jorda (2005) with adjustments proposed by Teulings and Zubanov (2014) are run to estimate the impulse response functions of GDP per capita to pandemics, as well as wars and weather conditions. Pandemics are found to have a prolonged and highly statistically significant effect on GDP per capita, with a pandemic resulting in the death of 1% of population increasing GDP per capita by a maximum of 0.3%,

this peak effect appearing after almost 20 years. To the contrary, the incidence of a war and one standard deviation increase in tree ring growth are estimated to depress GDP by about 1% and 0.5%, respectively, with the impacts dying out after 10-15 years.

The effects of pandemics are robust to various adjustments in the model specification and the use of GDP growth as a dependent variable. When a binary variable describing the incidence of pandemics is used as the main variable of interest, the sample is divided between Europe and the rest of the world, or the mostly deadly epidemics are excluded, the shape of impulse response functions remains similar, though quantitative estimates are larger and less precise.

Potential non-linearities with respect to pandemic size and its year of incidence are also investigated. It is found that small and largest pandemics have a positive effect on GDP per capita, while the impact of mid-size events is neutral or even negative. Similar conclusions are drawn from the specification that is non-linear with respect to time – early and recent pandemics (when losses in the population were largest and smallest, respectively) have a positive effect on GDP per capita, while epidemics in the 15th-17th centuries (which resulted in mid-size population losses) tend to decrease GDP.

**Related Literature**. This paper is related primarily to empirical studies on the economic consequences of pandemics. The most closely related paper is Jorda et al. (2020), in which authors use local projections to estimate the impact of pandemics (and wars) on interest rates and other macroeconomic variables, including GDP per capita, over a similar time frame. Their approach and data are much more limited, however – they do not use panel data, cover only the most (in)famous pandemics and wars, and neither control for weather conditions, include time trends, nor employ the Teulings-Zubanov adjustment. Finally, pandemics are represented by a binary variable. GDP per capita in the UK is found to be almost 10% higher 40 years after a pandemic, thus the effect is much larger than in this paper.

Focusing on more recent events and aiming at drawing conclusions relevant for the current COVID-19 crisis, Ma et al. (2020) use local projections and panel regressions to investigate the macroeconomic effects of the much smaller, but more recent post-World War 2 pandemics. They find that epidemics depress GDP by almost 3% at the outbreak and the effect does not fully die out within 5 years.

A couple of other papers study the short-run effects of the Spanish flu pandemic in given countries, investigating the relationship between regional pandemic incidence and economic outcomes (Brainerd and Siegler 2003, Karlsson et al. 2014, Barro et al. 2020). The results of these studies range from highly negative to positive impacts on GDP per capita.

Thus, the above-mentioned papers seem to suggest that the pre-20th century pandemics had a largely positive impact on GDP per capita, while the impact of more recent events (post-WW2 and to some extent the Spanish flu) was negative. This paper does not corroborate this conclusion, finding positive GDP per capita effects both in the medieval times and the 20th century.

The paper is also related to the literature on the pre-industrial economic growth in Europe, which debates whether per capita growth took place before 1750 (Jones 2000, Goldstone 2002, Broadberry and Wallis 2017, Goldstone 2019), in particular in England and the Netherlands (Broadberry 2013, de Pleijt and van Zanden 2019), pointing to the Black Death, overseas expansion and wars as potential sources of structural changes and growth (Pamuk 2007, Voigtlander and Voth 2013, de Pleijt and van Zanden 2016, O'Brien 2018, Jedwab et al. 2019, Prados de la Escosura and Rodriguez-Caballero 2020). Since the effects of pandemics are found to be prolonged and positive, the results of this paper support the view that pandemics played a role in spurring pre-industrial growth. The opposite is the case for wars, for which no long-run effects are identified.

The main contribution of this paper is thus a thorough study of both short-run and longerrun effects of pandemics on per capita income on a large sample of countries, controlling for wars and weather and using robust econometric techniques.

Regarding the current COVID-19 pandemic, the results of this paper do not point to declining positive effects of pandemics over time – if anything, the opposite is the case. Thus, while the results of this study should not be accepted at face value when relating to COVID-19 due to the obvious differences in the economic, organisational and healthcare structure with the pre-20th century world, they could be regarded as a reason of optimism that the long-run effects of the pandemic might not be as disastrous as often anticipated, and certainly not as disastrous as in the case of financial crises (Jorda et al. 2013, Reinhart and Rogoff 2014).

The paper is structured as follows: Section 2 describes the data, while Section 3 discusses the empirical framework. In Section 4, the baseline results and the results accounting for non-linearities with respect to pandemic size and time are presented, while Section 5 provides robustness checks. Finally, Section 6 discusses the limitations of the study and concludes.

## 2. Data

The dataset covers 33 countries<sup>2</sup> over the period of 1252 to 2016<sup>3</sup>. Only countries with continuous estimates of GDP per capita starting before World War I are included, so that the Spanish flu pandemic is covered<sup>4</sup>.

<sup>&</sup>lt;sup>2</sup> Argentina, Australia, Austria, Belgium, Canada, Switzerland, Chile, Cuba, Germany, Denmark, Spain, Finland, France, United Kingdom, Greece, Indonesia, India, Italy, Japan, Mexico, Malaysia, Netherlands, Norway, New Zealand, Peru, Philippines, Poland, Portugal, Romania, Russia, Sweden, United States, and South Africa.

<sup>&</sup>lt;sup>3</sup> Data on pandemics and wars is available over the whole 1252-2016 period, while the coverage of tree ring and GDP per capita data is limited. See the Appendix for details.

**Pandemics** are expressed in terms of the percentage of population that died due to a given event in a given year. Death toll estimates are more prevalent than infection estimates and are more likely to reflect the relative social and economic costs of an event. While these estimates might be imprecise and often vary across sources, they are preferable to a binary variable describing pandemic incidence that effectively gives equal weights to very different events. Nevertheless, the robustness of baseline results to the use of a binary dependent variable is discussed in subsection 5.3.

Data on incidence of pandemics and their death tolls is gathered from a large variety of sources – see Appendix A for more details. Death estimates are divided by population estimates from the Maddison Project Database (interpolated when necessary). Only pandemics that killed at least 0.1% of population per year are included. Endemic diseases are included only in the years of notable spikes in deaths. When pandemics are reported to have lasted for several years, the death toll is spread evenly over time, unless data suggests otherwise. The binary variable describing pandemic incidence, used in one of the robustness checks, additionally includes events that seem to be significant, but for which no death toll estimates are available.

Data on the size of pandemics is presented in Figure 1.

**GDP per capita data** comes primarily from the 2018 version of the Maddison Project Database (Bolt et al. 2018)<sup>5</sup>. As I am interested in GDP growth over time, rather than the comparison of relative GDP per capita levels across countries, the 2011 PPP benchmark series is used. Data for the Netherlands and the UK is projected backwards with estimates for Holland and England, respectively, assuming a constant relation of income levels between these regions and entire countries. Data for Italy is extended with estimates for North and Central Italy from Malanima (2011) in the same fashion. For Spain, recent estimates by Prados de la Escosura et al. (2020) are used. There are 9966 non-empty observations (out of 25245 possible ones for 33 countries over the whole time frame).

Since the incidence of pandemics is not necessarily fully exogenous, the inclusion of control variables might be warranted. In particular, historical evidence suggests that pandemics are often associated with <u>wars</u>. Looting, devastation, sieges, the resulting poor nutrition, as well as large concentrations and movements of troops are conducive to disease outbreaks and support their spread. For example, diseases are considered to be the main cause for the huge decline of population in Germany during the Thirty Years War (Outram 2001), plagues swept through Central and Eastern Europe during the Great Northern War (Kroll and Kruger 2006), typhus killed a large chunk of Napoleon's Grande Armée during the invasion

<sup>&</sup>lt;sup>4</sup> Not all such countries are included, however – Bolivia, Brazil, Colombia, Ecuador, Korea, Sri Lanka, Panama, Singapore, Taiwan, Uruguay and Venezuela all have continuous GDP per capita estimates starting before 1914 (though, with the exception of Venezuela, no earlier than 1870), but are not covered.

<sup>&</sup>lt;sup>5</sup> The 2020 edition of the database has been published recently, but it does not include important new time series. The main modification is the change back to the 1990 PPP benchmark, which does not have any effect on pre-1950 growth estimates and thus should not substantially influence the results of this paper.

of Russia (Conlon 2014), while one of the most deadly pandemics in history - the Spanish flu - broke out at the end of World War I.



Figure 1. Data on the size of the pandemics (% of population killed in a given year)

Source: Own compilation based on various sources – see Appendix A for details.

Thus, to control for a potential causal impact of wars on pandemics, data on war incidence, expressed in terms of a binary variable, is gathered from a very large number of sources – see Appendix B for details. Included are conflicts that either caused major destruction of physical capital and/or disruption of social and economic activity, or required a substantial military effort from a given country, i.e. a substantial part of its military force was used and actively engaged in warfare. For countries that did not exist as independent entities at the time of conflicts, they are reported to have taken part in a war if warfare had taken place on their territory or the population of the country had taken part in the conflict by being drafted into the military in significant numbers.

<u>Weather</u> is another factor that can potentially affect outbreaks and spread of pandemics. Long periods of drought may facilitate spread of diseases by causing poor nutrition and thus weaker immunity of both humans and disease transmitters. On the other hand, transmitters of some diseases (e.g. mosquitoes) benefit from wet weather conditions. Thus, long periods of drought followed by wet conditions might be particularly dangerous as they combine abundant populations of transmitters with weaker human immunity – some historical evidence suggests this could have been the reason why the 1545-48 cocoliztli epidemic in Mexico was so deadly (Acuna-Soto et al. 2002).

To proxy for the effects of weather, data on tree ring growth is used. Tree growth is clearly associated with weather conditions – in fact, temperature, precipitation and drought indices are reconstructed from tree ring data when more direct measures are not available (e.g. Briffa et al. 2004, Cook et al. 2004, Yang et al. 2014). At the same time, tree ring data is available for a larger number of countries over longer time periods than the reconstructed data on temperature, weather and droughts.

Tree ring data is gathered from the NOAA Paleoclimatology database and expressed in terms of the standardised growth index, computed according to the standard chronology method. When several locations are available for a given country, one of them is picked, taking into account time coverage and proximity in distance and altitude to most densely populated areas. For details on the data sources, see Appendix C. Tree ring data is not available for Cuba and Malaysia, shrinking the sample to 31 countries and 8218 observations.



Figure 2. Cross correlation plots

Cross-correlation plots (**Figure 2**) suggest that pandemics are indeed weakly, but statistically significantly and positively correlated with contemporaneous and past wars (up to 8 years back). The correlation between tree rings and pandemics is negative and strongest at 20-30 year lag, thus bad weather conditions seem to increase the likelihood of a pandemic outbreak. The relationship is not statistically significant, however.

## 3. Empirical framework

#### 3.1 Local projections and non-stationarity of GDP per capita

Local projections in the spirit of Jorda (2005) are used to estimate the impulse response function (IRF) of GDP per capita to pandemics (as well as wars and weather conditions) over the 40-year horizon. This method entails estimating a set of conditional forecast regressions for the IRF horizon h, which in the simplest case can be written as follows:

$$y_{t+h} = \alpha^h + \beta^h d_t + \varepsilon_t$$

where  $y_t$  is the dependent variable, and  $d_t$  is an exogenous shock for which the impulse response is computed. The impulse response function is made of the coefficients  $\beta^h$ :

$$IRF(h) = \beta^h$$

In the context of this paper, local projections have several advantages over IRFs computed from VAR models and analytical IRFs obtained from single equation ARDL models:

- Local projections are more robust to model misspecification, especially at longer horizons (Jorda 2005, Teulings and Zubanov 2014, Plagborg-Moller and Wolf 2020);
- Local projections are more flexible lags and non-linearities can be added in a straightforward manner<sup>6</sup>;
- Comparing to VAR models, local projections are much easier to estimate in a panel data setup.

However, Jorda (2005) local projections are biased if the shock reappears over the forecasting horizon (Teulings and Zubanov 2014). This can be corrected for by including shocks that take place between time t+1 and t+h in the regressions.

Moreover, if the variable of interest is not (fully) exogenous, control variables that are correlated with it (and the dependent variable) should be added to the regression in order to single out the exogenous component of the variation. However, even if the variable of interest is exogenous, the inclusion of control variables that are correlated with the dependent variable improves the efficiency of the model and IRF estimates (Jorda et al. 2020).

Non-stationarity of GDP per capita series is another issue relevant for the design of the empirical framework (**Figure 3**). Differencing would be the easiest solution, but it would make the study of the effects of pandemics on GDP per capita level more difficult (in particular, testing for statistical significance would be cumbersome). Given the aim of this paper to study longer-run impact of pandemics and the debatable quality of data on annual GDP growth rates in the earlier history, the effects on levels seem of more relevance than effects on growth rates.

<sup>&</sup>lt;sup>6</sup> Increasing the number of lags is limiting degrees of freedom much faster in VAR models than in local projections, while making the computation of IRFs much more cumbersome in ARDL models.

An alternative is to account for the structural shifts in trend GDP growth. With this aim, a common piecewise time trend is included in the regressions, with structural breaks in 1815 (the Congress of Vienna, i.e. the end of Napoleonic wars), 1945 (the end of World War II) and 1973 (the first oil shock). These dates reflect both important historical events and changes in the slope of log mean GDP per capita trend, as visible in **Figure 3**. Other specifications, including the regression using first differences, are considered in the robustness checks Section 5.



Figure 3. GDP per capita in the sample of 33 countries (2011 USD, log scale)

Source: Own calculations based on Maddison Project Database, extended with Malanima (2011) for Italy and Prados de la Escosura et al. (2020) for Spain.

#### 3.2 Baseline and non-linear specifications

All the above considered, the following set of regressions is estimated for h = 0, 1, ..., 40:

$$\begin{split} \ln y_{i,t+h} &= \mu_i^h + \sum_{j=0}^{30} \beta_j^h P_{i,t-j} + \sum_{k=1}^{30} \gamma_k^h y_{i,t-k} + \sum_{l=0}^{30} \theta_l^h war_{i,t-l} + \sum_{m=0}^{30} \vartheta_m^h tr_{i,t-m} + \delta_1^h t + \alpha_1^h vienna_t \\ &+ \delta_2^h vienna_t * t + \alpha_2^h ww2_t + \delta_3^h ww2_t * t + \alpha_3^h oil_t + \delta_4^h oil_t * t + \sum_{n=0}^{h-1} \pi_n^h P_{i,t+h-n} \\ &+ \sum_{o=0}^{h-1} \varphi_o^h war_{i,t+h-o} + \sum_{p=0}^{h-1} \omega_p^h tr_{i,t+h-p} + \varepsilon_{i,t+h} \end{split}$$

where  $y_{i,t}$  is ln GDP per capita for country *i* at time *t*,  $\mu_i$  is a country fixed effect,  $P_{i,t}$  is the percentage of people killed by a pandemic in country *i* at time *t*,  $war_{i,t}$  is a dummy variable equal to 1 if a conflict takes place in country *i* at time *t*,  $tr_{i,t}$  is the tree ring standardised growth index in country *i* at time *t*;  $vienna_t$ ,  $ww2_t$  and  $oil_t$  are dummies equal to 1 since 1815, 1945 and 1973, respectively, and 0 before; while  $\varepsilon_{i,t}$  is an idiosyncratic error term. The last 3 terms before the error term reflect the Teeulings-Zubanov adjustment, i.e. these are the shocks that take place between time t + 1 and t + h.

The impulse response of ln GDP per capita to a pandemic that killed 1% of the population is defined as follows:

$$IRF(h) = \beta_0^h$$

Impulse responses to wars and tree ring changes are constructed analogically, i.e.  $IRF_war(h) = \theta_0^h$  and  $IRF_tr(h) = \vartheta_0^h$ .

30 lags of GDP per capita as well as the pandemic, war and tree ring variables are included in the specification in order to control for the potential endogeneity of pandemics, and improve the model fit and thus estimate efficiency. Alternative numbers of lags are considered in the subsection 5.1 of the robustness checks.

The set of regressions is estimated with the fixed effects estimator. Even though this estimator is biased in a dynamic panel setting (Nickell 1981), the downward bias on the lagged dependent variable coefficient is diminishing with increasing time dimension of the panel, and thus can be safely ignored in the sample which covers 765 years.

Standard errors are corrected for heteroskedasticity and autocorrelation using Driscoll and Kraay (1998) approach. This is important since conditional forecasts of the type considered here suffer from serial correlation (Teulings and Zubanov 2014).

The baseline specification assumes that the effects of pandemics are linear in size. This might be a strong assumption, however – while small events are likely to have limited effects, especially in the longer run, large events might cause structural and institutional shifts, leading to pronounced effects in the long run.

Therefore, an alternative specification investigates whether the effects of pandemics on GDP per capita depend on the size of a pandemic. For this purpose, up to the 4th order polynomial terms of the pandemic variable at time *t* are added to the regressions (new terms in red):

$$\begin{split} \ln y_{i,t+h} &= \mu_i^h + \sum_{j=0}^{30} \beta_j^h P_{i,t-j} + \rho_1^h P_{i,t}^2 + \rho_2^h P_{i,t}^3 + \rho_3^h P_{i,t}^4 + \sum_{k=1}^{30} \gamma_k^h y_{i,t-k} + \sum_{l=0}^{30} \theta_l^h war_{i,t-l} \\ &+ \sum_{m=0}^{30} \vartheta_m^h tr_{i,t-m} + \delta_1^h t + \alpha_1^h vienna_t + \delta_2^h vienna_t * t + \alpha_2^h ww2_t + \delta_3^h ww2_t * t \\ &+ \alpha_3^h oil_t + \delta_4^h oil_t * t + \sum_{n=0}^{h-1} \pi_n^h P_{i,t+h-n} + \sum_{o=0}^{h-1} \varphi_o^h war_{i,t+h-o} + \sum_{p=0}^{h-1} \omega_p^h tr_{i,t+h-p} \\ &+ \varepsilon_{i,t+h} \end{split}$$

which results in the following expression for the impulse response function:

$$IRF(h) = \beta_0^h + 2\rho_1^h P_{i,t} + 3\rho_2^h P_{i,t}^2 + 4\rho_3^h P_{i,t}^3$$

Cubic form of the impulse response function gives enough flexibility to cover various forms of non-linearity.

The effects of pandemic might also differ over time with i.a. changing economic structures and healthcare practices. To test this hypothesis, the pandemic variable at time *t* is interacted with up to 2nd order polynomial terms of time (new terms in red):

$$\ln y_{i,t+h} = \mu_i^h + \sum_{j=0}^{30} \beta_j^h P_{i,t-j} + \tau_1^h P_{i,t} t + \tau_2^h P_{i,t} t^2 + \sum_{k=1}^{30} \gamma_k^h y_{i,t-k} + \sum_{l=0}^{30} \theta_l^h war_{i,t-l} + \sum_{m=0}^{30} \vartheta_m^h tr_{i,t-m} + \delta_1^h t + \alpha_1^h vienna_t + \delta_2^h vienna_t t + \alpha_2^h ww2_t + \delta_3^h ww2_t t + \alpha_3^h oil_t + \delta_4^h oil_t t + \sum_{n=0}^{h-1} \pi_n^h P_{i,t+h-n} + \sum_{o=0}^{h-1} \varphi_o^h war_{i,t+h-o} + \sum_{p=0}^{h-1} \omega_p^h tr_{i,t+h-p} + \varepsilon_{i,t+h}$$

which gives the following expression for the impulse response function:

$$IRF(h) = \beta_0^h + \tau_1^h t + \tau_2^h t^2$$

#### 4. Results

#### 4.1 Baseline estimates

The response of GDP per capita to a pandemic that kills 1% of the population in a given year, obtained from the baseline specification, is presented in **Figure 4**.

In the short run, positive and negative effects of pandemics on per capita income seem to roughly cancel each other out. After several years, positive effects associated with higher per capita land and capital begin to dominate. In the second decade following the pandemic, GDP per capita is about 0.25% higher, with the peak effect of 0.28% after 18 years. Afterwards, the effects of pandemics partially dissipate, but remain positive and statistically

significant even after 40 years. Thus, pandemics seem to have a positive impact on GDP per capita in the long run, perhaps because they induce positive institutional changes<sup>7</sup>.



**Figure 4.** Response of GDP per capita to a pandemic resulting in the death of 1% of the population in a given year

Shaded area is a 90% confidence band around response estimates.

The effects of pandemics are very different than those of war and weather (**Figure 5**). A war depresses GDP per capita by about 1% on impact and the effect gradually fades away over the next 10 years. Wars destroy capital while killing a smaller part of the population than pandemics. In line with the neoclassical growth model, this negative impact on capital labour ratio, together with other disruptive effects of wars, generates a negative response of GDP per capita. At the same time, wars – unlike pandemics - have no statistically significant effect on GDP per capita in the long run, contradicting the claims that they were one of the drivers of growth in pre-industrial Europe (Voigtlander and Voth 2013, O'Brien 2018).

In turn, one standard deviation increase in tree ring growth has a borderline significant, negative impact on GDP per capita that persists over the first decade, peaks at 0.5% after 11 years and fades away soon afterwards. These results are somewhat counterintuitive as weather conditions that are conducive to tree growth have a negative and quite persistent impact on per capita income. Perhaps weather conditions that support tree growth are not necessarily conducive to crop growth, especially in the case of excessively wet conditions.

<sup>&</sup>lt;sup>7</sup> For example, the Black Death is often mentioned as one of the driving factors behind the process of enclosure in England, which led to a more effective use of land and eventually the creation of a landless working class that found employment during the industrial revolution.

**Figure 5**. Response of GDP per capita to a war (left panel) and one standard deviation increase in tree ring index (right panel)



Shaded areas are 90% confidence bands around response estimates.

#### 4.2 Non-linearities and time dependence

The impulse responses obtained from specifications which allow for non-linearity with respect to the pandemic size and time are presented in **Figure 6**.

**Figure 6**. Response of GDP per capita to a pandemic: non-linear specifications with respect to the pandemic size (left panel) and time (right panel)



Impulse responses are scaled per 1% annual pandemic death toll.

Impulse responses vary wildly both with respect to the pandemic death toll and over time, and these differences are often counterintuitive and difficult to interpret. The smallest (<=2% of population killed) and largest (40% death toll) pandemics have a highly positive impact on GDP per capita in the first decade, slightly positive in the second decade and slightly negative afterwards. The effects of mid-sized pandemics are largely a mirror image – negative in the first decade and positive after more than 20 years. Especially a highly positive short-run impact of small pandemics is puzzling – those are unlikely to lead to large institutional changes, do not have large impact on capital labour ratios, and tend to happen

more recently, when more advanced containment measures are likely to negatively affect economic activity in the short run.

Time-varying impulse responses are consistent with the pandemic-size-varying ones. Even though the initial impact is – as intuition would suggest – negative, most recent pandemics tend to have a puzzlingly large positive effect on GDP per capita. Pandemics from the beginning of the sample (which coincides with the Black Death) also have a positive (though much smaller) effect on per capita income, while Renaissance time (1400-1600) pandemics tend to depress GDP per capita.

Due to the lack of a clear-cut pattern, the non-linear IRFs should be interpreted with extra caution. Having said that, one important conclusion may be drawn – there is no evidence that more recent and smaller pandemics have a less positive effect on GDP per capita than the baseline results would suggest. If anything, the opposite is true.

## 5. Robustness checks

### 5.1 Specification adjustments

As the first robustness check, the baseline specification is simplified by removing the piecewise time trend, the control variables and the Teulings-Zubanov adjustment (shocks that take place between time t+1 and t+h) as well as decreasing the number of lags of all the explanatory variables. The impulse response functions resulting from these robustness checks are presented in Figure 7.



**Figure 7.** Response of GDP per capita to a pandemic resulting in the death of 1% of the population in a given year: alternative model specifications

Estimates prove to be highly robust to the above-mentioned modifications. Dropping the time trends, the Teulings-Zubanov adjustments and decreasing the number of lags has very little effect on impulse responses. Only removing the control variables has a larger impact on the estimates, particularly in the longer run, when positive effects of pandemics no longer persist. Therefore, the inclusion of wars and tree rings is an important addition to the model, while other specification choices have virtually no impact on the conclusions of this study.

#### 5.2 GDP growth a dependent variable

Differencing is the alternative solution to the problem of non-stationarity of ln GDP per capita series. Growth of GDP per capita then becomes the dependent variable in the specification and the time trend variables are dropped. The dummy variables that account for time-varying equilibrium GDP growth rates are retained:

$$\begin{split} \hat{y}_{i,t+h} &= \mu_i^h + \sum_{j=0}^{30} \beta_j^h P_{i,t-j} + \sum_{k=1}^{30} \gamma_k^h \hat{y}_{i,t-k} + \sum_{l=0}^{30} \theta_l^h war_{i,t-l} + \sum_{m=0}^{30} \vartheta_m^h tr_{i,t-m} + \alpha_1^h vienna_t \\ &+ \alpha_2^h ww2_t + \alpha_3^h oil_t + \sum_{n=0}^{h-1} \pi_n^h P_{i,t+h-n} + \sum_{o=0}^{h-1} \varphi_o^h war_{i,t+h-o} + \sum_{p=0}^{h-1} \omega_p^h tr_{i,t+h-p} \\ &+ \varepsilon_{i,t+h} \end{split}$$

where  $\hat{y}_{i,t}$  is GDP per capita growth in country *i* at time *t*.

The impulse responses of GDP per capita growth and GDP per capita level to a pandemic are presented in **Figure** *8*.

**Figure 8**. GDP per capita growth as a dependent variable: the response of GDP per capita growth to a pandemic (left panel) and the cumulated response of GDP per capita level (right panel)



Response to a pandemic resulting in the death of 1% of the population in a given year. Shaded area is a 90% confidence band around response estimates.

The response of GDP per capita growth is volatile and difficult to interpret, but mostly positive in the first 18 years. The cumulated response is similar to the baseline estimate, but

points to a larger positive effect of pandemics in the longer run. Nevertheless, the conclusions remain the same. Thus, the results are robust to differencing GDP per capita.

#### 5.3 Binary pandemic variable

Like in Jorda et al. (2020), a binary variable can be used as the main variable of interest. In this way, one limits problems associated with the measurement error of an explanatory variable. It is also possible to cover pandemics for which death toll estimates are not available. The binary variable equals 1 only in the first year of an epidemic in a given country (in the pandemic size variable the death toll is spread evenly over the years of the pandemic).

The impulse response obtained from the binary pandemic variable specification is presented in **Figure 9**.



Shaded area is a 90% confidence band around response estimates.

The main qualitative conclusion remains the same – pandemics have a positive impact on GDP per capita. However, the effect peaks after 10 years instead of almost 20 and disappears after about 20 years, with no statistically significant impact in the longer run. Quantitatively, the effect is larger than in the baseline specification – GDP per capita is higher by as much as 3% 10 years after the pandemic<sup>8</sup>.

<sup>&</sup>lt;sup>8</sup> The numbers from the baseline and the binary variable specifications are not directly comparable, but taking into account that an average pandemic killed about 1.2% of population per year, and there are almost twice as few (169 to 308) pandemic events as observations with positive pandemic size (since only the first year of the pandemic is denoted in the binary variable case), it is clear that the binary variable specification points to significantly larger effects of pandemics than the baseline specification.

Relative to the baseline, the binary variable specification attributes higher weight to smaller events. It is not therefore surprising that the impulse response function is similar to those obtained from the non-linear specification for small pandemics (Figure 6, left panel). It also makes sense that no significant longer-run effect is found, as small events are unlikely to generate structural shifts.

#### 5.4 Excluding large pandemics

**Figure 1** clearly shows that while most of the pandemics are relatively small, killing at most a few per cent of the population, there have been several extremely large events such as the Black Death and the pandemics in the New World that dominate the sample and might heavily influence the results. Thus, these large events are eliminated from the sample to investigate whether the effects of pandemics differ across large and relatively small events<sup>9</sup>. The impulse response function without large pandemics is presented in **Figure 10**.



**Figure 10.** Specification without large pandemics: response of GDP per capita to a pandemic resulting in the death of 1% of the population in a given year

Shaded areas is a 90% confidence band around response estimates.

Qualitatively the results are similar to the baseline – in the first 20 years pandemics tend to have a positive impact on GDP per capita, though the response is not statistically significant at a 10% level. Peak effect comes much earlier (after 9 years), though, and clearly there is no positive impact in the longer run. The latter result is quite intuitive – only large events are

<sup>&</sup>lt;sup>9</sup> In particular, observations before 1500 are eliminated for European countries that experienced the Black Death and the recurring plague epidemics (Austria, Belgium, Switzerland, Germany, Denmark, Spain, France, the UK, Greece, Italy, Norway, the Netherlands, Poland, Portugal, Russia, and Sweden), observations before 1600 for countries that were colonised first (Cuba, Mexico, and Peru) and observations before 1650 for Chile and the US that were colonised later.

capable of generating structural shifts in the longer run. Quantitatively the effect is larger (1% at peak vs 0.3% in the baseline).

#### 5.5 Europe vs rest of the world

Do the effects of pandemics differ geographically? In particular, is there any difference between the Old and New Worlds? Acemoglu et al. (2001) claim that in the New World, diseases discouraged settlement and led to the creation of highly persistent, extractive institutions that limited future economic growth. Hence, the New World pandemics could as well have had negative impact on GDP per capita.

To test this hypothesis, the sample is divided between Europe (including Central and Eastern Europe; 18 countries) and the rest of the world (15 countries; all of them but Japan used to be colonies). The impulse responses to pandemics in these subsamples are shown in Figure 11.

**Figure 11**. Response of GDP per capita to a pandemic resulting in the death of 1% of the population in a given year: Europe (left panel) vs the rest of the world (right panel)



Shaded area are 90% confidence bands around response estimates.

The impulse response for Europe is virtually identical to the baseline, which is not very surprising given that almost 80% of the observations come from the Old Continent. In the rest of the world, the IRF is qualitatively similar – pandemics have a persistent, positive impact on GDP per capita. Thus, there is no evidence of the negative feedback loop between diseases and institutions as claimed by Acemoglu et al. (2001). The peak effect comes somewhat later (after 25 years) and is significantly larger (about 3%), though, and pandemics seem to have no impact in the longer run as the impulse response drops to virtually zero after 30 years.

## 6. Conclusions, discussion and caveats

The paper finds that pandemics have a positive impact on GDP per capita. The peak effect appears after 10 to 20 years, but positive impact persists for a few years more. In the short

run (the first few years) there is no statistically significant effect, however. These findings are robust to various specifications and subsamples.

In the baseline specification, the impulse response remains significant even after 40 years, thus pandemics have a positive impact on GDP per capita in the long run. However, this result disappears outside of Europe and when the binary pandemic variable is used, the control variables are dropped, or the largest pandemics are excluded. Judging from that, it seems to be the case that only large pandemics have a long-run effect on per capita income. This seems reasonable, but is not confirmed by the specification that explicitly accounts for non-linearities.

Quantitatively, in the baseline specification the effects are rather limited – at peak, GDP per capita increases by less than 0.3% following a pandemic that kills 1% of the population. However, quantitative results are not very robust - they tend to increase manifold when the binary pandemic variable is used and large pandemics are excluded, outside of Europe, or in the non-linear specifications.

The effects of pandemics differ from those of war and weather, the latter being proxied with tree rings. Both wars and higher tree ring growth depress GDP per capita in the first 10-15 years, with no long-run effects. Having said that, the negative effect of higher tree ring growth and its persistence 10-15 years afterwards is somewhat puzzling.

Economically, it seems that in the short run the positive impact of pandemics on land and capital per worker is offset by disruptions to economic activity and perhaps a rise in risk aversion. In the medium term, negative effects dissipate and the positive effects in the spirit of Malthus (higher per capita income following population decline) and the neoclassical growth model (higher land and capital per worker) begin to dominate. As population rebuilds itself through demographic compensation, these effects should ultimately dissipate as well. However, in some specifications, including baseline, positive impact of pandemics persists even after 40 years, particularly after large events. This suggests that large pandemics can lead to structural changes - in particular, they could make institutions more conducive to growth. The Black Death as one of the main drivers of enclosure in England is one such example.

Are the results of this paper in any way relevant in the context of the COVID-19 pandemic? COVID-19 significantly differs from the pandemics studied in this paper as its effects stem mostly from restrictions imposed on social and economic activity, while in the past the impacts were related mostly to population losses. Having said that, quarantines were used on a locally significant scale already during e.g. the Great Northern War of early 18th century, and have been growing increasingly important with advances in medical knowledge. Moreover, the specification non-linear with respect to time does not indicate that medium- and longer-run positive effects of pandemics decrease over time – the opposite is true, in fact – while the short-run effect is found to be negative, as expected.

Thus, while the studies focusing on post-WW2 pandemics (such as Ma et al. 2020) are clearly more relevant when one aims to study particularly the short-run effects of COVID-19, this paper could be regarded as an optimistic indication that unlike e.g. financial crises, the pandemic is unlikely to have negative effects in the longer run. Quite the contrary, these effects could be positive if COVID-19 speeds up innovation and structural shifts related to e.g. remote work, automation, or e-commerce.

There are numerous caveats to these results, however. First and foremost, there are many issues with the data. The pandemic death tolls are often no more than rough estimates, frequently limited to certain cities or regions. Even though the Black Death has been extensively studied and there is some relatively precise local data from parishes, estimates for the whole countries remain quite tentative. This is even more the case for the recurring plague epidemics or the first New World pandemics, while historical accounts tend to exaggerate death tolls. In the 17th-18th centuries the data and accounts are already more precise, but they are often confined to big cities. Thus, death toll estimates from this period are likely to be substantially biased downwards (think of e.g. the Great Plagues of Seville, London, Vienna, or Marseille).

Population estimates, to which the pandemic death tolls are related to obtain the main variable of interest, are also quite tentative prior to the 19th century. At the same time, they are quite sparse, therefore interpolation is often used to obtain a rough estimate for a given year. In the eras of wars and pandemics, these interpolations might be quite imprecise.

Other variables also suffer from data issues. In particular, GDP per capita estimates are quite tentative as well. They are constructed in the following way: at first, an estimate of GDP per capita is obtained for one or several specific years, using data on population size, its structure, urbanisation and incomes by social groups from one region. Later, these estimates are projected backwards and forwards using data on real wages from a major city and estimates of urbanisation, assuming that certain relations remain constant, to obtain a continuous GDP per capita series (see e.g. Malinowski and van Zanden 2017 for more details).

Thus, while GDP per capita estimates are relatively reliable for benchmark years (though still very imprecise compared to modern national accounts data), the in-between continuous estimates are very tentative. They are based on population interpolations, and hence do not account for large shifts in population due to pandemics and wars. Since in practice these are estimates for specific regions or even cities, they are influenced by local factors and might substantially diverge from the real values for the country as a whole, especially in the short run. Thus, GDP per capita estimates are most reliable in the longer run, while short run deviations should be interpreted with great caution.

Another problem with GDP per capita estimates is the constant price assumption. As relative prices change over time, which they are bound to do over such a long time frame, relative

levels of GDP per capita become biased, and this bias might be substantial (see Bolt and van Zanden 2020 for a discussion). However, this bias concerns per capita income comparisons across countries at a given point in time, while in this paper the focus is on GDP growth over time. Actually, differences in GDP per capita levels across countries are not even taken into account in the fixed effects estimation utilised in this paper. Therefore, the constant PPP bias is not a major issue here, though it might have some effect on impulse responses in the longer run, when changes in price structure are likely to become more significant.

The data on tree rings is plagued with similar issues as the GDP per capita data. These are estimates for a particular place, sometimes – when no better location is available – from not a very central region, not necessarily representative for a country as a whole. Moreover, observations are derived from various tree species, and thus are not entirely comparable between each other. The quality of the data is not constant over time, as early observations tend to be derived from fewer trees, leading to higher variation in tree ring growth. Finally, tree ring growth is not a perfect proxy for agricultural conditions, as weather conditions that are conducive to tree growth are not necessarily conducive also to crop growth.

Econometrically, all these data issues are likely to cause a measurement error in both the dependent and independent variables. As a result, parameter (IRF) estimates are likely to be biased towards zero and standard errors are likely to be inflated. Both issues make it more difficult to obtain statistically significant results. However, if the measurement error is not random – and for reasons presented above one might believe that in some cases it is not – then the results might as well be biased in another direction. Therefore, quantitative results should be treated with great caution.

Moving on from data issues, another important question is the stability of model parameters, particularly over time, which is related to the question whether the results of this study are relevant for the current policy debates. That the stability assumption is true seems highly doubtful, given the time span of the study, which includes evolving economic structures, improving knowledge about epidemic diseases and changing responses to them. While the stability assumption is eased in the non-linear specification, the results of this model specification are quite counterintuitive and difficult to interpret, potentially due to data issues. Hence, while there is some evidence that the main conclusions of this paper remain true even today, this result should be taken with more than a pinch of salt.

Omitted variables are another potential issue. The economic consequences of a pandemic might depend not only on its death toll, but also – as we have seen recently – on containment measures, whether it is a more local or nationwide event, infectiousness of the disease and its death rate, duration of the pandemic etc. There are also other variables that might be correlated with the likelihood of a pandemic outbreak and GDP per capita, e.g. hygiene practices, which are likely to have varied not only over time, but also across time, affecting pandemic outbreak and spread. Another such variable is trade linkages, which could have been conducive to pandemic spread, while boosting per capita income.

Going forward, more work is to be done on the data front, including the coverage of additional countries and a further selection of sources to make the data more comparable across time and countries. This relates not only to the pandemic variable, but also controls. It would also be desirable to scale the war variable in a similar fashion as the pandemic variable, either by casualties or the number of engaged troops. On the top of that, accounting for various other characteristics of pandemics other than the death toll, such as regional distribution and time scope, infectiousness, death rates, and containment measures, would definitely help to obtain more accurate and nuanced estimates of the economic consequences of pandemics.

At the same time, further investigation into non-linear effects of pandemics, both with respect to their size and time, and differences across countries in those effects, seems necessary to better understand the results obtained in this paper.

## References

Acemoglu, D., Johnson, S., & Robinson, J. A. (2001). *The Colonial Origins of Comparative Development: An Empirical Investigation*. American Economic Review, 91(5), 1369-1401.

Acuña-Soto, R., Stahle, D. W., Cleaveland, M. K., & Therrell, M. D. (2002). *Megadrought and Megadeath in 16<sup>th</sup> Century Mexico*. Emerging Infectious Diseases, 8(4), 360.

Barro, R. J., Ursúa, J. F., & Weng, J. (2020). The Coronavirus and the Great Influenza Pandemic: Lessons from the "Spanish Flu" for the Coronavirus's Potential Effects on Mortality and Economic Activity. NBER Working Paper, 26866.

Bolt, J., Inklaar, R., Jong, H. de, & Zanden, J. van (2018). *Rebasing 'Maddison': New Income Comparisons and the Shape of Long-run Economic Development*. GGDC Research Memorandum, 174.

Bolt, J., & Zanden, J. van (2020). *Maddison Style Estimates of the Evolution of the World Economy. A New 2020 Update*. Maddison-Project Working Paper, 15.

Brainerd, E., & Siegler, M. V. (2003). The Economic Effects of the 1918 Influenza Epidemic.

Briffa, K. R., Osborn, T. J., & Schweingruber, F. H. (2004). *Large-scale Temperature Inferences from Tree Rings: a Review*. Global and Planetary Change, 40(1-2), 11-26.

Broadberry, S. (2013). Accounting for the Great Divergence. Economic History Working Papers, 184/13.

Broadberry, S., & Wallis, J. J. (2017). *Growing, Shrinking, and Long Run Economic Performance: Historical Perspectives on Economic Development*. NBER Working Paper, 23343.

Conlon, J. M. (2014). The Historical Impact of Epidemic Typhus.

Cook, E. R., Woodhouse, C. A., Eakin, C. M., Meko, D. M., & Stahle, D. W. (2004). Long-term Aridity Changes in the Western United States. Science, 306(5698), 1015-1018.

Driscoll, J. C., & Kraay, A. C. (1998). *Consistent Covariance Matrix Estimation with Spatially Dependent Panel Data*. Review of Economics and Statistics, 80(4), 549-560.

Goldstone, J. A. (2002). *Efflorescences and Economic Growth in World History: Rethinking the "Rise of the West" and the Industrial Revolution*. Journal of World History, 323-389.

Goldstone, J.A. (2019). *Data and Dating the Great Divergence*, in T. Roy and G. Riello (eds.), *Global Economic History*, Bloomsbury, 38-53.

Jedwab, R., Johnson, N. D., & Koyama, M. (2019). *Pandemics, Places, and Populations: Evidence from the Black Death*. CESifo Working Paper, 7524.

Jones, E. L. (2000). Growth Recurring: Economic Change in World History. University of Michigan Press.

Jordà, Ò. (2005). *Estimation and Inference of Impulse Responses by Local Projections*. American Economic Review, 95(1), 161-182.

Jordà, Ò., Schularick, M., & Taylor, A. M. (2013). *When Credit Bites Back*. Journal of Money, Credit and Banking, 45(s2), 3-28.

Jordà, Ò., Singh, S. R., & Taylor, A. M. (2020). Longer-run Economic Consequences of Pandemics. NBER Working Paper, 26934.

Karlsson, M., Nilsson, T., & Pichler, S. (2014). *The Impact of the 1918 Spanish Flu Epidemic on Economic Performance in Sweden: An Investigation into the Consequences of an Extraordinary Mortality Shock*. Journal of Health Economics, 36, 1-19.

Kozlowski, J., Veldkamp, L., & Venkateswaran, V. (2020). *Scarring Body and Mind: the Long-term Belief-scarring Effects of Covid-19*. FRB St. Louis Working Paper, 2020-009.

Kroll, S., & Kruger, K. (2006). Städtesystem und Urbanisierung im Ostseeraum in der Frühen Neuzeit: urbane Lebensräume und historische Informationssysteme: Beiträge des wissenschaftlichen Kolloquiums in Rostock vom 15. Und 16. November 2004 (Vol. 12). LIT Verlag Münster.

Ma, C., Rogers, J., & Zhou, S. (2020). *Modern Pandemics: Recession and Recovery*. BOFIT Discussion Paper, 16-2020.

Malanima, P. (2011). *The Long Decline of a Leading Economy: GDP in Central and Northern Italy, 1300–1913.* European Review of Economic History, 15(2), 169-219.

Malinowski, M., & Zanden, J. L. van (2017). *Income and its Distribution in Preindustrial Poland*. Cliometrica, 11(3), 375-404.

Malthus, T. R. (1798). An Essay on the Principle of Population as It Affects the Future Improvement of Society, with Remarks on the Speculations of Mr Godwin, M. Condorcet, and other writers. J. Johnson.

Nickell, S. (1981). *Biases in Dynamic Models with Fixed Effects*. Econometrica: Journal of the Econometric Society, 1417-1426.

O'Brien, P. K. (2018). *The Costs and Benefits of Mercantilist Warfare*. Financial History Review, 25(1), 97-112.

Outram, Q. (2001). The Socio-economic Relations of Warfare and the Military Mortality Crises of the Thirty Years' War. Medical History, 45(2), 151-184.

Pamuk, Ş. (2007). *The Black Death and the Origins of the 'Great Divergence' across Europe*, 1300–1600. European Review of Economic History, 11(3), 289-317.

Plagborg-Møller, M., & Wolf, C. K. (2020). Local Projections and VARs Estimate the Same Impulse Responses.

Pleijt, A.M. de, & Zanden, J.L. van (2019). *Preindustrial Economic Growth, ca* 1270-1820, in C. Diebolt and M. Haupert (eds.), *Handbook of Cliometrics*, Springer Cham, 423-438.

Pleijt, A. M. de, & Zanden, J. L. van (2016). *Accounting for the "Little Divergence": What Drove Economic Growth in Pre-industrial Europe*, 1300–1800?. European Review of Economic History, 20(4), 387-409.

Prados de la Escosura, L., Álvarez-Nogal, C., & Santiago-Caballero, C. (2020). *Growth Recurring in Preindustrial Spain: Half A Millennium Perspective*. EHES Working Paper, 0177.

Prados De La Escosura, L., & Rodríguez-Caballero, C. V. (2020). *Growth, War, and Pandemics: Europe in the Very Long-run*. CEPR Discussion Paper, DP14816.

Reinhart, C. M., & Rogoff, K. S. (2014). *Recovery from Financial Crises: Evidence from 100 Episodes*. American Economic Review, 104(5), 50-55.

Schumpeter, J. (1942). Capitalism, Socialism and Democracy. Harper and Brothers.

Solow, R. M. (1956). A Contribution to the Theory of Economic Growth. The Quarterly Journal of Economics, 70(1), 65-94.

Teulings, C. N., & Zubanov, N. (2014). Is Economic Recovery a Myth? Robust Estimation of Impulse Responses. Journal of Applied Econometrics, 29(3), 497-514.

Voigtländer, N., & Voth, H. J. (2013). *The Three Horsemen of Riches: Plague, War, and Urbanization in Early Modern Europe.* Review of Economic Studies, 80(2), 774-811.

Yang, B., Qin, C., Wang, J., He, M., Melvin, T. M., Osborn, T. J., & Briffa, K. R. (2014). *A 3,500-Year Treering Record of Annual Precipitation on the Northeastern Tibetan Plateau*. Proceedings of the National Academy of Sciences, 111(8), 2903-2908.

## Appendix

## A. Pandemic data

Data on the incidence and the death toll of pandemics has been compiled from a large variety of sources, including – but not limited to – internet sources. When a range of or several death toll estimates are provided, the midpoint of the range/an average is calculated. When no death toll estimate is available, but a given epidemic is assessed to be significant, it is not included in the baseline pandemic variable, but it is covered by the binary variable measure. Death estimates are divided by population estimates from the Maddison Project Database (interpolated when necessary). Only pandemics estimated to have killed at least 0.1% of the population are included in the sample. Endemic diseases are included only in the years of notable spikes in deaths. When pandemics are reported to have lasted for several years, the death toll is spread evenly over time, unless data suggests otherwise.

Table A. 1 lists data sources by the types of pandemics.

Pandemic/disease	Data sources				
The Black Death	"Black Death", "Black Death migration", "Bubonic plague", "List of epidemics",				
and the recurring	"Pandemic" and "Second plague pandemic" Wikipedia articles; Gould and Pyle				
plague epidemics	(1896), Payne (1911), Biraben (1976), Horrox (1994), Austin Alchon (2003),				
(Second plague	Benedictow (2004), Bray (2004), Hays (2005), Deleo and Hinnebusch (2005), Hays				
pandemic)	(2009), Cohn (2010), Gottfried (2010), Hatcher (2010), Welford and Bossak (2010),				
	Byrne (2012), Aberth (2013), Ziegler (2013), Cesana et al. (2017), Karlsson (2020).				
	Austria: "Great Plague of Vienna" Wikipedia article.				
	France: "Great Plague of Marseille" Wikipedia article, Jones (1996), Parker				
	(2001), Duchene and Contrucci (2004).				
	Germany: Eckert (1978).				
	Italy: "1629-1631 Italian plague" and "Naples Plague" Wikipedia articles,				
	Wyman (1897), Beloch (1937), Morrison et al. (1985), Scasciamacchia et al. (2012),				
	Tognotti (2013), White (2014).				
	Romania: "Caragea's plague" and "Great Plague of 1738" Wikipedia articles,				
	Ionescu (1974).				
	Russia: Collins (1671), Melikishvili (2006).				
	Spain: "Great Plague of Seville" Wikipedia article, Payne (1973).				
	Sweden: "Black Death in Sweden" Wikipedia article, Alexander (2002), Myrd				
	(2003), Griffiths (2009).				
	UK: "1563 London plague", "1592-1593 London plague", "Black Death in				
	England" and "The Great Plague of London" Wikipedia articles; Nichols (1823),				
	Creighton (1891), Russell (1948), Appleby (1980), Goldberg (1996), Moote and				
	Moote (2004), Porter (2009), Newman (2012), Lewis (2016).				
Great Northern	"Great Northern War plague outbreak" Wikipedia article, Lorinser (1837),				
War plague	Sticker (1908), Helleiner (1967), Bohn (1989), bei der Wieden (1999), Kossert				
outbreak (a part of	(2005), Kroll and Kruger (2006), Vuorinen (2007), Burchardt et al. (2009),				
the second plague	Frandsen (2010), Gose (2009).				
pandemic)					
Sweating sickness	"Sweating sickness" Wikipedia article.				

Tumbuc	"List of anidomics" and "Pandomic" Willingdia article Pattorson (1002) Conlan					
турниз	(2014)					
	(2014).					
	(1026) Canada: 184/ North American typnus epidemic <sup>®</sup> Wikipedia article, Gallagher					
	(1936).					
	<b>Poland:</b> Szukata (2020).					
Early flu pandemics	"1510 influenza pandemic" and "1557 influenza pandemic" Wikipedia articl					
	Bergeron (1872), Morens et al. (2010).					
New World	"List of epidemics" and "Pandemic" Wikipedia articles.					
pandemics	Australia: Cumpston (1914).					
	Canada: "1837 Great Plains smallpox epidemic" Wikipedia article, Desjardins					
	(1996).					
	Mexico: "Cocoliztli epidemics" Wikipedia article, Acuna-Soto et al. (2000),					
	Acuna-Soto et al. (2002).					
	<b>Peru</b> : Cook (2004), Kiracofe and Marr (2009).					
	US: "1721 Boston smallpox outbreak", "1775-1782 North American smallpox					
	epidemic", "1793 Philadelphia yellow fever epidemic", "1837 Great Plains					
	smallpox epidemic" and "Massachusetts smallpox epidemic" Wikipedia articles;					
	Carey (1793), Thatcher (1828), Kelly and Burrage (1920), Krebsbach (1996),					
	Houston and Houston (2000), Marr and Cathey (2010), Daschuk (2013), Purvis					
	(2014).					
Yellow fever	"List of epidemics" and "Yellow fever in Buenos Aires" Wikipedia articles,					
	Chisholm (1911).					
Cholera	"Cholera outbreaks and pandemics", "List of epidemics" and "Pandemic"					
	Wikipedia articles; Hays (2005), Byrne (2008), Rosenberg (2009).					
	First pandemic: "1817-1824 cholera pandemic" Wikipedia article, Moreau de					
	Jonnes (1831), McGrew (1960), Arnold (1993).					
	Second pandemic: "1826-1837 cholera pandemic" Wikipedia article, Raymond					
	Durand (1980), Beardslee (2000).					
	Third pandemic: "1846-1860 cholera pandemic", "1853 Copenhagen cholera					
	outbreak" and "Nashville, Tennessee cholera epidemic (1849–1850)" Wikipedia					
	articles; Bowling (1866), Unruh (1993), Sugihara et al. (1996), Hosking (2001),					
	Kohn (2008).					
	Fourth pandemic: "1863-1875 cholera pandemic" Wikipedia article, Barua and					
	Greenough (1992), Smallman-Raynor and Cliff (2004).					
	<b>Fifth pandemic</b> : "1881-1896 cholera pandemic" Wikipedia article.					
	Sixth pandemic: "1899-1923 cholera pandemic" Wikipedia article.					
	Seventh nandemic <sup>•</sup> "1961–1975 cholera pandemic <sup>•</sup> Wikipedia article. Kotar and					
	Gessler (2014).					
Malaria	"Groningen epidemic" Wikipedia article					
Third plaque	"Bombay plague epidemic" and "Third plague pandemic" Wikipedia articles					
nandemic	Low (1899) Low (1902) Fager (1908)					
Russian flu	"1889-1890 pandemic" Wikipedia article Parsons (1891) Mouritz (1921) Ryan					
Russiun ju	(2008) Charles River Editors (2020)					
Dolio	"List of anidomics" Wikingdia article Ochman and Roser (2017)					
Tono Smanich flu	"Spanish fly" Wilcipedia article, US Conque Bureau (1020), Warron (1021), Jordan					
эритып ји	(1927) Aman (1990) Patterson and Pule (1901) Mayor (2000) Johnson and					
	(1727), $(1770)$ , $(1770)$ , $(1770)$ , $(1770)$ , $(1771)$ , $(1771)$ , $(1770)$ , $(200)$ , $(2000)$ , $(2000)$ , $(200$					
	(2010) Enkoroka (2010) Killingrou and Dhilling (2011) Chardren et al. (2009), Carbonetti					
	(2010), Erkoreka (2010), Killingray and Phillips (2011), Chandra et al. (2012),					
	Kadusin (2012), Chowell et al. (2014), Yang et al. (2014), Grabowski et al. (2017),					
	Spinney (2018), Spreeuwenberg et al. (2018), Arnold (2019), Mata (2020).					
Asian flu	"1957–1958 influenza pandemic" Wikipedia article, Clark (2008), Jackson (2009),					

	Viboud et al. (2016), Nickol and Kindrachuk (2019), Kutzner (2020).		
Hong Kong flu	"Hong Kong flu" Wikipedia article.		
Smallpox in India	"1974 smallpox epidemic in India" Wikipedia article.		

## B. War data

Data on the incidence of wars has been gathered mostly from the very rich internet sources – Wikipedia lists of wars ("List of wars: 1000-1499", "List of wars: 1500-1799", "List of wars: 1800-1899", "List of wars: 1900-1944", "List of wars: 1945-1989", "List of wars: 1990-2002", "List of wars: 2003-present"), country-specific Wikipedia lists of wars ("List of wars involving Argentina", "List of wars involving Australia", "List of wars involving Austria", "List of wars involving Belgium", "List of wars involving Canada", "List of wars involving Switzerland", "List of wars involving Chile", "List of wars involving Cuba", "List of wars involving France"), all the Wikipedia articles linked there, and references therein. The second major source is the Conflict Catalog (Brecke 1999).

Figure A. 1 Data on the incidence of wars



Source: Own compilation based on various sources.

Included are conflicts that are judged to have caused major destruction of physical capital and/or disruption of social and economic activity, or have required a substantial military effort from a given country, i.e. a substantial part of its military force was used and actively

engaged in warfare (rather than only deployed to an area of conflict). For countries that did not exist as independent entities at the time of conflicts, they are reported to have taken part in a war if warfare had taken place on their territory or the population of the country had taken part in the conflict (by being drafted into the military in significant numbers).

Figure A. 1 presents the data on the incidence of wars.

#### C. Tree ring data

Tree ring data is gathered from the NOAA Paleoclimatology database and expressed in terms of the standardised growth index, computed according to the standard chronology method. When several locations are available for a given country, one of them is picked, taking into account time coverage and proximity in distance and altitude to most densely populated areas. When data is not available for a given country, a proximate location from a neighbouring country is chosen. For Cuba and Malaysia this is not possible, though, and thus the tree ring data covers only 31 countries.

Tree ring data details are shown in Table A. 2 and the data itself is presented in Figure A. 2.





Source: Own compilation based on the sources from the NOAA Paleoclimatology database (for details see Table A.2).

Country	Author/study	Location	Altitu	Tree type	Time coverage
			de		
Argentina	Villalba	El Arrasayal	880 m	JGAU	1766-1985
Australia	Lamarche	Bruny Island	380 m	PHAS	1542-1975
Austria	Giertz	Obergurgl	2000 m	LADE	1604-1972
Belgium	Hoffsummer	Meuse Valley	0 m	QUSP	1252-1989
Canada	Archambault and Bergeron	Lac Duparquet	274 m	THOC	1252-1987
Switzerland	Schweingruber	Krauchtal BE	550 m	PISY	1714-1976
Chile	Holmes	Caramavida	900 m	ARAR	1440-1972
Germany	Billamboz	Bodensee	450 m	QUSP	1275-1986
Denmark	Schweingruber	Gotland, Sweden	50 m	PISY	1252-1987
Spain	Genova Fuster, Fernandez-Cancio and Perez Antelo	Torreton	1500 m	PINI	1485-1988
Finland	Eronen	Lieksa Koivujoki	150 m	PISY	1588-1983
France	Lambert, Lavier and Trenard	Bourgogne 29 Master	0 m	QUSP	1252-1991
United	Wilson et al.	Southern-Central	45-185		
Kingdom	(2013)	England	m	QUSP	1252-2009
Greece	Kuniholm	Chalkidiki Arnaia Barbara	600 m	QUFR	1740-1979
Indonesia	D'Arrigo, Krusic, Jacoby and Buckley	Bigin, Java	75 m	TEGR	1839-1995
India	Borgaonkar, Pant, and Rupa Kumar	Narkhanda	3000 m	ABPI	1590-1989
Italy	Schweingruber	Sierra de Crispo	2000 m	PILE	1441-1980
Japan	Davi, D'Arrigo, Jacoby, Buckley and Kobayashi	Mount Asahidake, Hokkaido	1350 m	PCGN	1532-1997
Mexico	Stahle et al. (2011)	Barranca de Amealco	1970 m	Montezuma baldcypress	1252-2008
Netherlands	Jansma and van Rijn (1252-1457), Jansma (1458- 1650), Maessen (1783-1990)	Maastricht St. Jan's Church (1252-1457), Oegstgeest (1458- 1650), whole Netherlands (1783-1990)	0-20 m	QUSP (1252- 1650), PISY (1783-1990)	1252-1650, 1783-1990
Norway	Kirchhefer	Forfjorddalen 2	110 m	PISY	1252-1994
New Zealand	Ahmed, Boswijk and Ogden	Manaia Sanctuary	350 m	AGAU	1269-1998
Peru	Lopez et al. (2017)	Purubi, Bolivia	446 m	CEMC	1798-2010
Philippines	Cook et al. (2010)	Bakun	-	PIKE	1721-2005
Poland	Wazny	East Pomerania	20 m	QURO	1252-1985
Portugal	Shestakova et al. (2019)	Pinar de Lillo, Spain	1600 m	PISY	1511-2002
Romania	Schweingruber	Novaci	1650 m	РСАВ	1804-1981
Russia	Schweingruber	Nyuchpas	160 m	LASI	1649-1991

## Table A. 2 Tree ring data details

Sweden	Schweingruber	Gotland	50 m	PISY	1252-1987
United States	Stahle et al. (2013)	Average of Blackwater River and Devil's Gut	-	Baldcypress	1252-1993
South Africa	Lamarche and Dunwiddie	Die Boss	1330 m	WICE	1564-1976

#### D. GDP data

GDP per capita data comes primarily from the 2018 version of the Maddison Project Database (Bolt et al. 2018)<sup>10</sup>. As I am interested in GDP growth over time, rather than the comparison of relative GDP per capita levels across countries, the 2011 PPP benchmark series (rgdpnapc) is used. Data for the Netherlands and the UK is projected backwards with estimates for Holland and England, respectively, assuming a constant relation of income levels between these regions and whole countries. Data for Italy is extended with estimates for North and Central Italy from Malanima (2011) in the same fashion. For Spain, recent estimates by Prados de la Escosura et al. (2020) are used.





Source: Maddison Project Database (2018 edition), Malanima (2011), Prados de la Escosura et al. (2020).

<sup>&</sup>lt;sup>10</sup> The 2020 edition of the database has been published recently, but it does not include important new time series. The main modification is the change back to the 1990 PPP benchmark, which does not have any effect on pre-1950 growth estimates and thus should not substantially influence the results of this paper.

GDP per capita data is presented in Figure A. 3.

#### **E.** References

Aberth, J. (2013). From the Brink of the Apocalypse: Confronting Famine, War, Plague and Death in the Later *Middle Ages*. Routledge.

Acuña-Soto, R., Romero, L. C., & Maguire, J. H. (2000). *Large Epidemics of Hemorrhagic Fevers in Mexico* 1545-1815. The American Journal of Tropical Medicine and Hygiene, 62(6), 733-739.

Acuña-Soto, R., Stahle, D. W., Cleaveland, M. K., & Therrell, M. D. (2002). *Megadrought and Megadeath in 16<sup>th</sup> Century Mexico*. Emerging Infectious Diseases, 8(4), 360.

Alchon, S. A. (2003). A Pest in the Land: New World Epidemics in a Global Perspective. UNM Press.

Alexander, J. T. (2002). *Bubonic Plague in Early Modern Russia: Public Health and Urban Disaster*. Oxford University Press.

Åman, M. (1990). *Spanska sjukan: den svenska epidemin 1918-1920 och dess internationella bakgrund*. Studia Historica Upsaliensia, 160.

Ansart, S., Pelat, C., Boelle, P. Y., Carrat, F., Flahault, A., & Valleron, A. J. (2009). *Mortality Burden of the 1918–1919 Influenza Pandemic in Europe*. Influenza and Other Respiratory Viruses, 3(3), 99-106.

Appleby, A. B. (1980). *Epidemics and Famine in the Little Ice Age*. The Journal of Interdisciplinary History, 10(4), 643-663.

Arnold, D. (1993). *Colonizing the Body: State Medicine and Epidemic Disease in Nineteenth-Century India*. University of California Press.

Arnold, D. (2019). *Dearth and the Modern Empire: The 1918–19 Influenza Epidemic in India*. Transactions of the Royal Historical Society, 29, 181-200.

Barry, J. M. (2004). *The Site of Origin of the 1918 Influenza Pandemic and its Public Health Implications*. Journal of Translational Medicine, 2(1), 1-4.

Barua, D., & Greenough III, W. B. (1992). Cholera: Current Topics in Infectious Disease. Springer.

Beardslee, G. W. (2000). The 1832 Cholera Epidemic in New York State. The Early America Review, 3(2).

bei der Wieden, B. (1999). *Die Entwicklung der Pommerschen Bevölkerung*, 1701 *bis* 1918. Veröffentlichungen der Historischen Kommission für Pommern (vol. 33). Forschungen zur Pommerschen Geschichte (vol. 5). Böhlau.

Beloch, K. J. (1937). Bevölkerungsgeschichte Italiens: Grundlagen: die Bevölkerung Siziliens und des Königreichs Neapel. De Gruyter.

Benedictow, O. J. (2004). The Black Death, 1346-1353: The Complete History. Boydell & Brewer.

Bergeron, G. (1872). Des Caractères Généraux des Affections Catarrhales Aiguës. Adrien-Delahaye.

Biraben, J. N. (1976). Les Hommes et la Peste en France et Dans les Pays Européens et Méditerranéens, Les Hommes Face à la Peste: Civilisations et Sociétés. Mouton.

Bohn, R. (1989). *Das Handelshaus Donner in Visby und der gotländische Außenhandel im 18. Jahrhundert*. Quellen und Darstellungen zur hansischen Geschichte. 33. Böhlau.

Bolt, J., Inklaar, R., Jong, H. de, & Zanden, J. van (2018). *Rebasing 'Maddison': New Income Comparisons and the Shape of Long-run Economic Development*. GGDC Research Memorandum, 174.

Bowling, W. K. (1866). *Cholera, as it Appeared in Nashville in 1849, 1850, 1854 and 1866*. University Book and job Office, Medical College, WHF Ligon.

Bray, R. S. (2004). Armies of Pestilence: The Impact of Disease on History. James Clarke & Co.

Brecke, P. (1999). *Violent Conflicts 1400 AD to the Present in Different Regions of the World*. 1999 Meeting of the Peace Science Society.

Burchardt, J., Meissner, R. K., & Burchardt, D. (2009). *Oddech Śmierci – Zaraza Dżumy w Wielkopolsce i w Poznaniu w Pierwszej Połowie XVIII wieku*. Nowiny Lekarskie, 78(1), 79-84.

Byrne, J. P. (2008). Encyclopedia of Pestilence, Pandemics and Plagues. Greenwood Press.

Byrne, J. P. (2012). Encyclopedia of the Black Death. ABC-CLIO.

Carbonetti, A. (2010). *Historia de una Epidemia Olvidada: La Pandemia de Gripe Española en la Argentina, 1918-1919*. Desacatos, (32), 159-174.

Carey, M. (1793). A Short Account of the Malignant Fever, Lately Prevalent in Philadelphia: with a Statement of the Proceedings that Took Place on the Subject in Different Parts of the United States.

Cesana, D., Benedictow, O. J., & Bianucci, R. (2017). *The Origin and Early Spread of the Black Death in Italy: First Evidence of Plague Victims from 14th-century Liguria (Northern Italy).* Anthropological Science, 125(1), 15-24.

Chandra, S., Kuljanin, G., & Wray, J. (2012). Mortality from the Influenza Pandemic of 1918–1919: the Case of India. Demography, 49(3), 857-865.

Chisholm, H., ed. (1911). *Yellow Fever*. In: *Encyclopædia Britannica*. 28 (11th ed.). Cambridge University Press, pp. 910–911.

Chowell, G., Simonsen, L., Flores, J., Miller, M. A., & Viboud, C. (2014). *Death Patterns during the 1918 Influenza Pandemic in Chile*. Emerging Infectious Diseases, 20(11), 1803.

Clark, W. R. (2008). Bracing for Armageddon?: The Science and Politics of Bioterrorism in America. Oxford University Press.

Cohn, S. K. (2010). Black Death, Social and Economic Impact of the. In: Bjork, R. E. (ed.). The Oxford Dictionary of the Middle Ages. Oxford University Press.

Collins, S. (1671). The Present State of Russia. Newman.

Conlon, J. M. (2014). The Historical Impact of Epidemic Typhus.

Cook, N. D. (2004). Demographic Collapse: Indian Peru, 1520-1620. Cambridge University Press.

Cook, E. R., Anchukaitis, K. J., Buckley, B. M., D'Arrigo, R. D., Jacoby, G. C., & Wright, W. E. (2010). *Asian Monsoon Failure and Megadrought during the Last Millennium*. Science, 328(5977), 486-489.

Creighton, C. (1891). A History of Epidemics in Britain, From AD 964 to the Extinction of Plague. Cambridge University Press.

Cumpston, J. H. L. (1914). The History of Small-pox in Australia, 1788-1908. AJ Mullett.

Daschuk, J. W. (2013). *Clearing the Plains: Disease, Politics of Starvation, and the Loss of Aboriginal Life.* University of Regina Press.

DeLeo, F. R., & Hinnebusch, B. J. (2005). A Plague upon the Phagocytes. Nature Medicine, 11(9), 927-928.

Desjardins, B. (1996). *Demographic Aspects of the* 1702-1703 *Smallpox Epidemic in the St. Lawrence Valley*. Canadian Studies in Population, 49-67.

Duchêne, R., & Contrucci, J. (2004). Marseille: 2600 ans d'histoire. Fayard.

Eckert, E. A. (1978). Boundary Formation and Diffusion of Plague: Swiss Epidemics from 1562 to 1669. In: *Annales de démographie historique* (pp. 49-80). Société de Démographie Historique.

Erkoreka, A. (2010). *The Spanish Influenza Pandemic in Occidental Europe (1918–1920) and Victim Age.* Influenza and Other Respiratory Viruses, 4(2), 81-89.

Frandsen, K. E. (2010). The Last Plague in the Baltic Region 1709-1713. Museum Tusculanum Press.

Gallagher, J. A. (1936). *The Irish Emigration of 1847 and Its Canadian Consequences*. CCHA Report, 3, 43-57.

Goldberg, J. (1996). *Introduction*. In: Ormrod, M. & Lindley, P.G. (ed.). *The Black Death in England*. Stamford: Paul Watkins. pp. 1–15.

Göse, F. (2009). Prenzlau im Zeitalter des 'Absolutismus' (1648–1806). In Neitmann, K., & Schich, W. (eds.). Geschichte der Stadt Prenzlau. Geiger. pp. 140–184.

Gottfried, R. S. (2010). Black Death. Simon and Schuster.

Gould, G. M., & Pyle, W. L. (1896). Anomalies and Curiosities of Medicine. The Julian Press, Inc.

Grabowski, M. L., Kosińska, B., Knap, J. P., & Brydak, L. B. (2017). *The Lethal Spanish Influenza Pandemic in Poland*. Medical Science Monitor: International Medical Journal of Experimental and Clinical Research, 23, 4880.

Griffiths, T. (2009). Stockholm: A Cultural History. Oxford University Press.

Hatcher, J. (2010). The Black Death: An Intimate History. Hachette UK.

Hays, J. N. (2005). Epidemics and Pandemics: their Impacts on Human History. ABC-CLIO.

Hays, J. N. (2009). *The Burdens of Disease: Epidemics and Human Response in Western History*. Rutgers University Press.

Helleiner, K. F. (1967). *The Population of Europe from the Black Death to the Eve of the Vital Revolution*. The Cambridge Economic History of Europe, 4, 1-96.

Horrox, R. (Ed.). (1994). *Fifteenth-century Attitudes: Perceptions of Society in Late Medieval England*. Cambridge University Press.

Hosking, G. A. (2001). Russia and the Russians: A History. Harvard University Press.

Houston, C. S., & Houston, S. (2000). *The First Smallpox Epidemic on the Canadian Plains: in the Furtraders' Words.* Canadian Journal of Infectious Diseases, 11.

Ionescu, Ş. X. (1974). Bucureștii în Vremea Fanarioților. Ed. Dacia.

Jackson, C. (2009). *History Lessons: the Asian Flu Pandemic*. British Journal of General Practice, 59(565), 622-623.

Johnson, N. P., & Mueller, J. (2002). *Updating the Accounts: Global Mortality of the 1918-1920 "Spanish" Influenza Pandemic.* Bulletin of the History of Medicine, 105-115.

Jones, C. (1996). Plague and its Metaphors in Early Modern France. Representations, 53, 97-127.

Jordan, E. O. (1927). Epidemic Influenza. A Survey. American Medical Association.

Karlsson, G. (2020). Iceland's 1100 years: History of a Marginal Society. Oxford University Press.

Kelly, H. A., & Burrage, W. L. (1920). American Medical Biographies. Norman, Remington Company.

Killingray, D. & Phillips, H. (Eds.). (2011). *The Spanish Influenza Pandemic of 1918-1919: New Perspectives*. Routledge Studies in the Social History of Medicine, 12.

Kiracofe, J. B., & Marr, J. S. (2009). Marching to Disaster: The Catastrophic Convergence of Inca Imperial Policy, Sand Flies, and El Nino in the 1524 Andean Epidemic. In: Sandweiss, D. H., & Quilter, J. El Niño, Catastrophism, and Culture Change in Ancient America. Dumbarton Oaks Research Library and Collection, 145-165.

Knobler, S. L., Mack, A., Mahmoud, A., & Lemon, S. M. (2005). *The Threat of Pandemic Influenza: Are We Ready? Workshop Summary*. Institute of Medicine (US) Forum on Microbial Threats. National Academies Press (US).

Kohn, G. C. (2008). Encyclopedia of Plague and Pestilence: from Ancient Times to the Present. Infobase Publishing.

Kossert, A. (2005). Die Große Pest 1709–1711. In: Ostpreußen. Geschichte und Mythos. Siedler. pp. 96–103.

Kotar, S. L., & Gessler, J. E. (2014). Cholera: A Worldwide History. McFarland.

Krebsbach, S. (1996). *The Great Charlestown Smallpox Epidemic of 1760*. The South Carolina Historical Magazine, 97(1), 30-37.

Kroll, S., & Kruger, K. (2006). Städtesystem und Urbanisierung im Ostseeraum in der Frühen Neuzeit: urbane Lebensräume und historische Informationssysteme: Beiträge des wissenschaftlichen Kolloquiums in Rostock vom 15. Und 16. November 2004 (Vol. 12). LIT Verlag Münster.

Kutzner, M. (2020). *Debatte zur Herkunft der Asiatischen Grippe 1957*. Deutschland Archiv. Federal Agency for Civic Education.

Lewis, C. (2016). Disaster Recovery: New Archaeological Evidence for the Long-term Impact of the 'Calamitous' Fourteenth Century. Antiquity, 90(351), 777-797.

Lopez, L., Stahle, D., Villalba, R., Torbenson, M., Feng, S., & Cook, E. (2017). *Tree Ring Reconstructed Rainfall over the Southern Amazon Basin*. Geophysical Research Letters, 44(14), 7410-7418.

Lorinser, C. I. (1837). Die Pest des Orients: Wie Sie Entsteht und Verhütet Wird. Enslin.

Malanima, P. (2011). *The Long Decline of a Leading Economy: GDP in Central and Northern Italy, 1300–1913.* European Review of Economic History, 15(2), 169-219.

Marr, J. S., & Cathey, J. T. (2010). New Hypothesis for Cause of Epidemic Among Native Americans, New England, 1616–1619. Emerging Infectious Diseases, 16(2), 281.

Mata, S. (2020). 1917 American Pandemic: The Origin of the Spanish Flu. Independently published.

Mayor, S. (2000). Flu Experts Warn of Need for Pandemic Plans. BMJ. 321 (7265): 852.

McGrew, R. E. (1960). *The First Cholera Epidemic and Social History*. Bulletin of the History of Medicine, 34(1), 61-73.

Melikishvili, A. (2006). *Genesis of the Anti-plague System: The Tsarist Period*. Critical Reviews in Microbiology, 32(1), 19-31.

Moote, A. L., & Moote, D. C. (2004). The Great Plague: the Story of London's Most Deadly Year. JHU Press.

Moreau de Jonnès, A. (1831). Rapport au Conseil Supérieur de Santé Sur le Choléra-morbus Pestilentiel. D'elhoungne.

Morens, D. M., North, M., & Taubenberger, J. K. (2010). *Eyewitness Accounts of the 1510 Influenza Pandemic in Europe*. The Lancet, 376(9756), 1894-1895.

Morrison, A. S., Kirshner, J., & Molho, A. (1985). *Epidemics in Renaissance Florence*. American Journal of Public Health, 75(5), 528-535.

Myrdal, J. (2003). *Digerdöden, Pestvågor och ödeläggelse: ett Perspektiv på Senmedeltidens Sverige*. Sällsk. Runica et Mediævalia.

Newman, K. L. (2012). *Shutt Up: Bubonic Plague and Quarantine in Early Modern England*. Journal of Social History, 45(3), 809-834.

Nichols, J. (1823). The Progresses and Public Processions of Queen Elizabeth: Among which are Interspersed Other Solemnities, Public Expenditures, and Remarkable Events During the Reign of that Illustrious Princess. Collected from Original Manuscripts, Scarce Pamphlets, Corporation Records, Parochial Registers, Etc., Etc.; Illustrated, with Historical Notes. J. Nichols and Son.

Nickol, M. E., & Kindrachuk, J. (2019). A Year of Terror and a Century of Reflection: Perspectives on the Great Influenza Pandemic of 1918–1919. BMC Infectious Diseases, 19(1), 117.

Parker, G. (2001). Europe in crisis, 1598-1648. Wiley-Blackwell.

Patterson, K. D., & Pyle, G. F. (1991). *The Geography and Mortality of the 1918 Influenza Pandemic*. Bulletin of the History of Medicine, 65(1), 4-21.

Patterson, K. D. (1993). Typhus and its Control in Russia, 1870–1940. Medical History, 37(4), 361-381.

Payne, J. F. (1911). *Plague*. In: Chisholm, H. (ed.). *Encyclopædia Britannica*. 11th ed. Cambridge University Press.

Payne, S. G. (1973). A History of Spain and Portugal. Madison: University of Wisconsin Press.

Porter, S. (2009). The Great Plague. Amberley Publishing.

Prados de la Escosura, L., Álvarez-Nogal, C., & Santiago-Caballero, C. (2020). *Growth Recurring in Preindustrial Spain: Half A Millennium Perspective*. EHES Working Paper, 0177.

Purvis, T. L. (2014). Colonial America to 1763. Infobase Publishing.

Radusin, M. (2012). *The Spanish Flu, Part II: The Second and Third wave*. Vojnosanitetski Pregled, 69(10), 917-927.

Raymond Durand, L. M. (1980). *Depesze z powstańczej Warszawy 1830-1831 : raporty konsula francuskiego w Królestwie Polskim*. Czytelnik.

Rosenberg, C. E. (2009). The Cholera Years: The United States in 1832, 1849, and 1866. University of Chicago Press.

Russell, J. C. (1948). British Medieval Population. Albuquerque, U. of New Mexico P.

Scasciamacchia, S., Serrecchia, L., Giangrossi, L., Garofolo, G., Balestrucci, A., Sammartino, G., & Fasanella, A. (2012). *Plague Epidemic in the Kingdom of Naples*, *1656–1658*. Emerging Infectious Diseases, 18(1), 186.

Shestakova, T. A., Voltas, J., Saurer, M., Berninger, F., Esper, J., Andreu-Hayles, L., ... & Masson-Delmotte, V. (2019). *Spatio-temporal Patterns of Tree Growth as Related to Carbon Isotope Fractionation in European Forests under Changing Climate*. Global Ecology and Biogeography, 28(9), 1295-1309.

Smallman-Raynor, M. R., & Cliff, A. D. (2004). Impact of Infectious Diseases on War. Infectious Disease Clinics, 18(2), 341-368.

Spinney, L. (2018). Pale Rider: The Spanish Flu of 1918 and How It Changed the World. Vintage.

Spreeuwenberg, P., Kroneman, M., & Paget, J. (2018). *Reassessing the Global Mortality Burden of the 1918 Influenza Pandemic*. American Journal of Epidemiology, 187(12), 2561-2567.

Stahle, D. K., Burnette, D. J., & Stahle, D. W. (2013). *A Moisture Balance Reconstruction for the Drainage Basin of Albemarle Sound, North Carolina*. Estuaries and Coasts, 36(6), 1340-1353.

Stahle, D. W., Diaz, J. V., Burnette, D. J., Paredes, J. C., Heim, R. R., Fye, F. K., ... & Stahle, D. K. (2011). *Major Mesoamerican Droughts of the Past Millennium*. Geophysical Research Letters, 38(5).

Sticker, G. (1908). Die Pest. Toepelmann.

Sugihara, K., Robb, P. G., & Yanagisawa, H. (Eds.) (1996). Local Agrarian Societies in Colonial India: Japanese Perspectives. Psychology Press.

Szukała, Ł. (2020). *Epidemie na ziemiach polskich*. Nauka w Polsce, <u>https://naukawpolsce.pap.pl/aktualnosci/news%2C81227%2Cepidemie-na-ziemiach-polskich.html</u>

Thatcher, J. (1828). American Medical Biography, or, Memoirs of Eminent Physicians Who Have Flourished in America.

Tognotti, E. (2013). Lessons from the History of Quarantine, from Plague to Influenza A. Emerging Infectious Diseases, 19(2), 254.

United States Census Bureau (1920). *Mortality Statistics 1918: Nineteenth Annual Report.* Government Printing Office.

Unruh, J. D. (1993). *The Plains Across: The Overland Emigrants and the Trans-Mississippi West, 1840-60.* University of Illinois Press.

Vuorinen, H. S. (2007). *History of plague epidemics in Finland. In* Signoli, M., Cheve, D., Adalian, P., Boetsch, G., & Dutour, O. (eds.). *Peste: Entre épidemies et Sociétés.* Firenze University Press. pp. 53–56.

Vaughan, W. T. (1921). Influenza: an Epidemiologic Study. American Journal of Hygiene.

Viboud, C., Simonsen, L., Fuentes, R., Flores, J., Miller, M. A., & Chowell, G. (2016). *Global Mortality Impact of the 1957–1959 Influenza Pandemic*. The Journal of Infectious Diseases, 213(5), 738-745.

Welford, M., & Bossak, B. H. (2010). *Revisiting the Medieval Black Death of 1347–1351: Spatiotemporal Dynamics Suggestive of an Alternate Causation.* Geography Compass, 4(6), 561-575.

White, A. (2014). Plague and Pleasure: The Renaissance World of Pius II. CUA Press.

Wilson, R., Miles, D., Loader, N. J., Melvin, T., Cunningham, L., Cooper, R., & Briffa, K. (2013). *A Millennial Long March–July Precipitation Reconstruction for Southern-Central England*. Climate Dynamics, 40(3-4), 997-1017.

Wyman, W. (1897). The Black Plague. The North American Review, 164(485), 441-452.

Yang, W., Petkova, E., & Shaman, J. (2014). *The 1918 Influenza Pandemic in New York City: Age-specific Timing, Mortality, and Transmission Dynamics.* Influenza and Other Respiratory Viruses, 8(2), 177-188.

Ziegler, P. (2013). The Black Death. Faber & Faber.